

Citation: Tobin RJ, Harrison LE, Tully MK, Lubis IND, Noviyanti R, Anstey NM, et al. (2024) Updating estimates of *Plasmodium knowlesi* malaria risk in response to changing land use patterns across Southeast Asia. PLoS Negl Trop Dis 18(1): e0011570. https://doi.org/10.1371/ journal.pntd.0011570

Editor: Richard Reithinger, RTI International, UNITED STATES

Received: August 3, 2023

Accepted: January 16, 2024

Published: January 22, 2024

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0011570

Copyright: © 2024 Tobin et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Prediction results for each bootstrapped model, rasters of summary statistics, the code used to produce results, and

RESEARCH ARTICLE

Updating estimates of *Plasmodium knowlesi* malaria risk in response to changing land use patterns across Southeast Asia

Ruarai J. Tobin¹*, Lucinda E. Harrison², Meg K. Tully¹, Inke N. D. Lubis³, Rintis Noviyanti⁴, Nicholas M. Anstey⁵, Giri S. Rajahram⁶, Matthew J. Grigg⁵, Jennifer A. Flegg², David J. Price^{1,7}, Freya M. Shearer^{1,8}*

 Infectious Disease Dynamics Unit, Melbourne School of Population and Global Health, The University of Melbourne, Melbourne, Australia, 2 School of Mathematics and Statistics, The University of Melbourne, Melbourne, Australia, 3 Department of Paediatrics, Faculty of Medicine, Universitas Sumatera Utara, Medan, Indonesia, 4 Eijkman Research Center for Molecular Biology, BRIN, Jakarta, Indonesia, 5 Menzies School of Health Research and Charles Darwin University, Darwin, Australia, 6 Infectious Diseases Society Kota Kinabalu Sabah, Menzies School of Health Research, Clinical Research Unit, Hospital Queen Elizabeth II, and Clinical Research Centre, Queen Elizabeth Hospital, Ministry of Health, Kota Kinabalu, Malaysia,
7 Doherty Institute for Infection and Immunity, The Royal Melbourne Hospital and The University of Melbourne, Melbourne, Australia, 8 Infectious Disease Ecology and Modelling Group, Telethon Kids Institute, Perth, Australia

* ruarai.tobin@unimelb.edu.au (RJT); freya.shearer@unimelb.edu.au (FMS)

Abstract

Background

Plasmodium knowlesi is a zoonotic parasite that causes malaria in humans. The pathogen has a natural host reservoir in certain macaque species and is transmitted to humans via mosquitoes of the *Anopheles* Leucosphyrus Group. The risk of human *P. knowlesi* infection varies across Southeast Asia and is dependent upon environmental factors. Understanding this geographic variation in risk is important both for enabling appropriate diagnosis and treatment of the disease and for improving the planning and evaluation of malaria elimination. However, the data available on *P. knowlesi* occurrence are biased towards regions with greater surveillance and sampling effort. Predicting the spatial variation in risk factors and account for spatial bias in detection.

Methods & results

We extend and apply an environmental niche modelling framework as implemented by a previous mapping study of *P. knowlesi* transmission risk which included data up to 2015. We reviewed the literature from October 2015 through to March 2020 and identified 264 new records of *P. knowlesi*, with a total of 524 occurrences included in the current study following consolidation with the 2015 study. The modelling framework used in the 2015 study was extended, with changes including the addition of new covariates to capture the effect of deforestation and urbanisation on *P. knowlesi* transmission.

the updated occurrence database have been made available at osf.io/k5bsa (DOI <u>10.17605/OSF.IO/</u> K5BSA).

Funding: This study was supported through funding provided by the Australian Centre for International Agricultural Research (ACIAR), as part of the 'Evaluating zoonotic malaria transmission and agricultural and forestry land use in Indonesia' (ZOOMAL) project (LS/2019/116, www.aciar.gov.au). Further support for this project was provided by the National Health and Medical Research Council of Australia through its Centres of Research Excellence (ACREME, GNT1134989 www.nhmrc.gov.au). FMS was supported by the National Health and Medical Research Council of Australia Investigator Grant Scheme (Emerging Leader Fellowship, 2021/GNT2010051 www. nhmrc.gov.au). JAF was supported by the Australian Research Council (ARC, FT210100034 and DP200100747 www.arc.gov.au). LEH was supported by a Melbourne Research Scholarship from the University of Melbourne (www.unimelb. edu.au). MJG was supported by the National Health and Medical Research Council of Australia Investigator Grant Scheme (Emerging Leader 2 Fellowship, 2023/GNT2017436 www.nhmrc.gov. au). GSR and MJG were supported by the National Institutes of Health, USA (R01AI160457-01 www. nih.gov). GSR was also supported by the Malaysian Ministry of Health (Grant Number BP00500/117/1002 www.moh.gov.my). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Discussion

Our map of *P. knowlesi* relative transmission suitability estimates that the risk posed by the pathogen is highest in Malaysia and Indonesia, with localised areas of high risk also predicted in the Greater Mekong Subregion, The Philippines and Northeast India. These results highlight areas of priority for *P. knowlesi* surveillance and prospective sampling to address the challenge the disease poses to malaria elimination planning.

Author summary

Plasmodium knowlesi is a parasite that can cause malaria when it infects humans. Although most people do not experience severe illness from *Plasmodium knowlesi* infection, a small number will develop serious or even fatal disease. The parasite is found naturally in some monkeys throughout Southeast Asia, and spreads from these monkeys to humans through mosquitoes. Previous research predicted where the risk of being infected is highest according to what we know about the environment across Southeast Asia, such as if there are forests in an area or if the altitude is high. In this work, we extend this previous research with more up-to-date data on environmental conditions and infections to predict the risk of being infected with *Plasmodium knowlesi*. We show that the risk *Plasmodium knowlesi* poses to humans is high across much of Southeast Asia, and that the disease will continue to challenge national goals to eliminate malaria.

Introduction

Plasmodium knowlesi is a zoonotic pathogen of growing public health concern in Southeast Asia. The pathogen has a reservoir in *Macaca fascicularis* and the closely related *Macaca nemestrina* and *Macaca leonina* macaques, and is transmitted between macaques and from macaques to humans via mosquito vectors of the *Anopheles* Leucosphyrus Group [1,2]. Although demonstrated experimentally [3], evidence of direct human-to-human transmission of *P. knowlesi* occurring in nature is limited [1,4–7].

Infection by *P. knowlesi* most often causes mild to moderate illness in humans [8]. However, a range of outcomes are possible, with both asymptomatic infection [9–11] and severe disease being reported. Studies of patients presenting to health care facilities in Malaysia reported severe disease in around 6–9% of patients [12,13].

For humans, the likelihood of contracting a *P. knowlesi* infection has been found to be dependent upon a range of risk factors, with case-control and seroprevalence studies demonstrating associations between environmental variables and the occurrence of infection. A seroprevalence survey performed in Malaysia and the Philippines found that prior infection with *P. knowlesi* was associated with the proximity of forested areas to an individual's home and the clearing of forest near their home [14]. A similar study performed in northern Sabah, Malaysia, found associations between prior infection and an individual reporting that they have had activity in forested areas or that they have had contact with macaques [15]. A population-based case-control study performed within Sabah, Malaysia, found an association between current *P. knowlesi* infection and an individual reporting either that they had recently cleared vegetation or that their home was in proximity to long grass [11].

The spatial epidemiology of *P. knowlesi* malaria has historically been poorly understood. This is partially due to widespread misdiagnosis. By clinical presentation, the symptoms of *P*. *knowlesi* infection can be easily misattributed to other major human species of malaria such as *P. vivax* or *P. falciparum* [16]. Under microscopic examination, the parasite appears almost identical to *P. malariae* [17] and the early ring stages of *P. falciparum* [18]. One review of historical microscopy diagnoses demonstrated that across 375 studies, 57% of *P. knowlesi* infections were misdiagnosed [17]. In addition to misdiagnosis, the understood spatial distribution of *P. knowlesi* malaria has been biased by differences in surveillance effort. Within peerreviewed literature, reported *P. knowlesi* infections are most common in Malaysia, which is likely reflective of both high burden and a substantial surveillance effort in the country [19]. Indigenous cases of *P. knowlesi* malaria have also been detected in Brunei, Cambodia, Indonesia, Laos, Myanmar, the Philippines, Thailand, and Vietnam, but these have historically been the result of small-scale prospective sampling efforts and individual case reports. One study reported identifying *P. knowlesi* in India within the Andaman and Nicobar Islands [20].

The elimination of malaria in at least 20 countries by 2025 is listed as a key milestone of the World Health Organisation's 2016–2030 Global technical strategy for malaria [21]. P. knowlesi presents a challenge to these efforts, since interventions that are effective against the human malaria species such as indoor residual spraying will be less effective against P. knowlesi due to the pathogen's persistence in wildlife reservoirs. Furthermore, cross-reactivity of antibodies between P. knowlesi and the closely genetically related P. vivax may provide protection against P. knowlesi infection [22], implying that the elimination of P. vivax in a region could lead to reduced immunity and subsequently an increase in the number of P. knowlesi infections [19].

The incidence of *P. knowlesi* in humans appears to be increasing within Southeast Asia; in Malaysia, the number of recorded human *P. knowlesi* infections doubled over the period from 2015 to 2018 [23]. A similar trend is visible in the rising number of case reports within Indonesia [24]. Though these trends may simply reflect improvements in surveillance [23], it has been suggested that deforestation in the region may be leading to a real increase in the number of human *P. knowlesi* infections [24–27]. A primary driver of deforestation in the region is the development of oil palm or timber plantations, which produce an environment that is believed to be of enhanced risk for *P. knowlesi* infection, with plantation labourers being required to live and work in proximity to recently disturbed forests that may contain *P. knowlesi* reservoirs and vectors.

Sustained transmission of vector-borne zoonoses can only occur at the nidus where pathogen, host and vector are present in sufficient abundance [28]. For each of these, certain constraints limit their distribution, for example: a pathogen may be unable to survive at certain temperatures; a host may be displaced by human activity; and a vector may be unable to reproduce without access to standing water. The field of geospatial information systems (GIS) provides a large amount of data on such environmental and anthropological factors [29]. Environmental niche modelling utilises this geospatial data to identify relationships between the presence of a pathogen, host or vector and the environments in which they have been observed, allowing for prediction of the suitability for transmission of a vector-borne zoonosis such as *P. knowlesi* across a geographic area of interest [30].

In 2015, Shearer and colleagues applied a niche modelling approach to produce the first predictive map of *P. knowlesi* malaria risk across Southeast Asia [31]. This map provided an initial evidence base for identifying areas where disease surveillance and epidemiological investigations would be most informative to improve understanding of *P. knowlesi* malaria risk. Since the publication of the 2015 occurrence database and risk map, the volume of *P. knowlesi* data has increased across Southeast Asia, with this including detections of the pathogen in new locations. As new data accrues, it is important to update risk predictions to ensure that the most up-to-date evidence is available to public health researchers, practitioners, and policymakers. Furthermore, since 2015, studies providing evidence of the importance of

deforestation in the risk of *P. knowlesi* malaria have been published, and novel datasets characterising spatial and temporal variation in land use patterns have become available.

In this study, we present updates to the *P. knowlesi* infection database and risk map produced in 2015 [31]. We perform a comprehensive review of the literature from October 2015 through to March 2020 to produce a consolidated database of *P. knowlesi* infection occurrences across Southeast Asia. By combining this occurrence dataset with data on a range of environmental covariates using a niche modelling framework, we produce updated predictions of relative suitability for *P. knowlesi* transmission to humans at fine-scale across Southeast Asia. We compare the outputs of our model to those from the 2015 model.

Methods

Infection data

The infection occurrence database is a listing of reported locations of *P. knowlesi* infections in either humans, macaques or mosquitoes. The infection occurrences used in the 2015 analysis were extracted from literature published up to October 2015. In order to identify new occurrences, we searched the 'Web of Science' database on March 2nd 2020, using the keywords "knowlesi" *or* "monkey malaria" and filtered for results published after October 2015 (Fig 1A). Following the exclusion of laboratory studies, we extracted infection occurrence records from publications which utilised validated *P. knowlesi*-specific diagnostics (i.e. semi-nested PCR or a combination of microscopy and molecular techniques, as in the 2015 review [31]). The data collection protocol used was the same as in the 2015 analysis and further detail can be found therein [31]. We combined the collected infection occurrences produced by the current study (n = 264, Fig 1) with those identified in the 2015 analysis (n = 260).

Each location in the infection occurrence database could either take the form of a point or a polygon record. We created point records where the likely exposure site was reported with enough precision that it could be assigned to a 5×5 km grid cell. Where this level of precision was not available, we created *polygonal* records, assigning the likely exposure site to a region bounded by a polygon (Fig 1B). We created these polygons as either administrative level 1 (the first subdivision below national, e.g. state or province) or administrative level 2 (the second subdivision below national, e.g. district or regency) then disaggregated these polygons onto to the 5×5 km grid for model fitting, prediction and evaluation.

Prior to model fitting and evaluation, we excluded nine records which spanned an area greater than 1,000 grid cells (approximately 25,000 km²). These records were unlikely to affect results given that each had substantial overlap with other more precise spatial records.

Covariate data

The infection risk model incorporated 20 environmental covariates (Table 1), each a 5×5 km gridded raster covering Southeast Asia. Of these 20 covariates, we treated 14 as time-varying with an annual resolution, allowing the model to associate each infection occurrence records with covariate values corresponding to the year the infection was recorded, capturing the variation of risk factors over time. Data for these annually-varying covariates were available for each year from 2001 to 2019, extending upon the coverage of the 2015 model (which covered 2001 to 2015). We assigned five samples which were collected before the year 2001 covariate values for the year 2001.

While tasseled-cap values (transformed Landsat imagery which can help differentiate areas of vegetation and urbanisation) and human population density were included as synoptic (static) variables in the 2015 model, in this work we incorporated them as temporally-varying covariates. The 2015 model incorporated an urban accessibility metric which defined the travel



Fig 1. The results of the March 2020 literature review. A: Study and sample selection process for the 2020 infection occurrence database. Records were produced via a literature review which was performed on March 2nd 2020, filtering for publications released after October 2015. B: Newly extracted point and polygon occurrence records across Southeast Asia by spatial type. Admin 1 regions are the first subdivision below national, e.g. state or province. Admin 2 regions are the second subdivision below national, e.g. district or regency. C: The number of occurrence samples in each occurrence database by the year the sample was collected. Administrative boundary base maps sourced from the Malaria Atlas Project (CC BY 3.0, [32]) and international boundaries from the US Department of State Large Scale International Boundaries dataset (public domain, [33]).

https://doi.org/10.1371/journal.pntd.0011570.g001

time to the nearest city of 50,000 people or more by land- or water-based travel in the year 2000 [34]. Here, we instead used the healthcare accessibility surface—a modelled measure of travel time to the nearest healthcare facility produced by the Malaria Atlas Project which used data up to mid-2019 [35]—as a measure of urban accessibility.

We replaced the intact and disturbed forest coverage layers used in the 2015 model with covariates that better captured the temporal and spatial dynamics of forest change in Southeast Asia. The forest coverage data sets used in the 2015 model were derived from the Intact Forest Landscapes project, which utilised a strict, manually assessed criteria for defining intact versus disturbed forest [31,36,37]. However, the temporal resolution of this dataset is low, with data only available for four distinct years (2000, 2013, 2016 and 2020). We chose instead to utilise data provided through the Global Forest Change project, which provides annual data on tree coverage over the last 20 years on forest presence at the resolution of 1 arc-second (roughly 30 m) [38].

We aggregated the Global Forest Change dataset up to the 5×5 km grid over the Southeast Asia study region through the calculation of both a tree coverage and a tree loss metric. We defined tree loss to be the proportion of forest area lost within each 5×5 km cell for each study year. Similarly, we defined tree coverage as the proportion of land where forest coverage was present at the beginning of the Global Forest Change data period and where no Table 1. The set of raster covariate datasets used in model fitting and prediction. Differences in raster datasets between this work and those used in the 2015 P. knowlesi risk model appear in bold. STRM: Shuttle Radar Topography Mission, MODIS: Moderate Resolution Imaging Spectroradiometer, IGBP: International Geosphere-Biosphere Programme.

| Name | Description | Temporal resolution |
|------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------|---------------------|
| Macaca fascicularis suitability | Modelled suitability of inhabitation by macaques of species <i>M. fascicularis</i> [37]. | Synoptic |
| Macaca nemestrina suitability | Modelled suitability of inhabitation by macaques of species <i>M. nemestrina</i> [37]. | Synoptic |
| Anopheles Leucosphyrus Group suitability | Modelled suitability of inhabitation by mosquitoes of the Anopheles Leucosphyrus Group [37]. | Synoptic |
| SRTM elevation | Mean elevation [39] | Synoptic |
| Tasseled cap wetness s.d. | Tasseled-cap transformed MODIS data [40, 41]. Now treated as temporally-varying. | Annual |
| Tasseled cap wetness mean | """ | Annual |
| Tasseled cap brightness s.d. | "" | Annual |
| <i>Plasmodium falciparum</i> temperature suitability | Modelled temperature suitability index for <i>P. falciparum</i> transmission used as proxy for suitability of <i>P. knowlesi</i> [42]. | Synoptic |
| Forest loss | Proportion of land where forest coverage has been lost in a given year [38]. Replaced the disturbed forest dataset . | Annual |
| Forest coverage | Proportion of land with forest coverage present in a given year [38]. Replaced the intact forest dataset . | Annual |
| Healthcare accessibility | Modelled duration travel time to the nearest healthcare facility [35]. Replaced the urban accessibility dataset . | Synoptic |
| WorldPop human population | Mean human population density [43, 44]. Now treated as temporally-varying. | Annual |
| Open shrublands | Proportion of land with given land classification [45]. | Annual |
| Woody savannas | " " " | Annual |
| Savannas | " " " | Annual |
| Grasslands | " " " | Annual |
| Permanent wetlands | " " " | Annual |
| Croplands | " " " | Annual |
| Cropland/natural vegetation mosaic | « « « « | Annual |
| Urban and built up | α α α | Annual |

https://doi.org/10.1371/journal.pntd.0011570.t001

subsequent loss was recorded up until each study year. As the Global Forest Change project has not calculated forest gain past the year 2012, we were not able to include any possible increase in forest coverage.

Model fitting

We utilised a bootstrapped boosted regression tree modelling framework to characterise relationships between a region's environment and the occurrence of *P. knowlesi* transmission. Regression trees produce an approximation of some latent function (e.g. the probability of a *P. knowlesi* infection occurring) by recursively splitting across potential predictor variables (e.g. environmental covariates). The points at which these splits occur and the value assigned across each split region are selected such that the error between the regression tree and the observations is minimised [46]. Boosted regression trees extend upon the regression tree framework by producing a large number of trees and combining them in an ensemble (a process known as boosting) such that they better approximate the latent function [47]. Boosted regression trees are able to fit complex nonlinear responses including high-dimensional interactions between explanatory variables due to their hierarchical tree structure and have been shown to exhibit high predictive accuracy [48]. Finally, bootstrapping of the boosted regression tree process can be performed, allowing for uncertainty in the output to be estimated [49]. When applied to presence-absence data (such as from a systematic survey), niche models generally use a binomial likelihood to represent the probability of a species being present at a given location. Where most of the data available for modelling are presence-only, as is the case for *P. knowlesi* malaria, it is common practice in niche modelling to supplement occurrence records with "background" points to represent areas where the species or disease has not been reported [48]. A variety of approaches have been employed to select background points, including sampling to ensure that their spatial distribution emulates the sampling bias in the presence records [50].

Most *P. knowlesi* occurrences to date are recorded in Malaysia, Brunei and Singapore, with all three of these countries having eliminated the human malaria species (*e.g.*, *P. vivax* and *P. falciparum*), such that that *P. knowlesi* is routinely considered a potential cause of malaria cases. Outside of these countries, surveillance for *P. knowlesi* is limited and infection records are sparse. As per the 2015 study, the goal of our niche modelling analysis is to predict broadly into the under-sampled regions outside of Malaysia, Brunei, and Singapore, using a model fit to data from within these three countries (i.e. the model training region, Fig 2A) where we can account for reporting bias through the selection of background points. Data from outside of these three countries formed the evaluation dataset (i.e. the model evaluation region, Fig 2B), which we used to assess the model's predictive ability outside of the training region.

Background points were produced as in the 2015 analysis [31]. To produce background points for the human records, we sampled points across the training region weighted by human population density [43], under the assumption that human *P. knowlesi* infections would be more likely to be detected within more populous areas. Background points for mosquito records were similarly produced with sampling weighted by human population density, under the assumption that the locations of mosquito infection studies would be selected based on the presence of human *P. knowlesi* cases. To produce macaque background records, we sampled points from a survey of macaques and other mammals [37], as we expected this survey to have similar sampling bias to that of macaque *P. knowlesi* infection records. As in the 2015 model [31], the geographic distribution of *Macaca leonina*—a putative host species of *P. knowlesi* which was only classified as a species distinct from *Macaca nemestrina* in 2001 [51]—has not been included as an explanatory covariate in model fitting as the species is not found in the model training region.

To produce each bootstrap we performed sampling with replacement across each of the combined occurrence polygons, occurrence point records and background points, using occurrence records present in the training region of Malaysia, Brunei and Singapore. We constrained this sampling so that at least 10 presence and 10 background points were present within each bootstrap. Each polygon occurrence record was then reduced to a single point location; this was achieved by selecting a point at random uniformly across each polygon for each bootstrap. As areas of overlapping polygons therefore have a greater probability of a point being sampled, we present the density of overlapping polygons in Fig 2A. For each bootstrap, we assigned weights to sampled points such that the sum of weights for presence points was equal to the sum of weights for the background points, and environmental values were assigned to each point from the set of covariate rasters corresponding to the spatial location and year the sample was recorded. We produced a covariate for host species, indicating if the sample was collected from a human, a mosquito or a macaque. We repeated this process to produce 500 bootstrapped datasets.

For each bootstrapped dataset, we fit boosted regression trees using the gbm3 and seegSDM packages. Hyperparameters for model fitting were unchanged from the defaults provided by seegSDM version 0.1-9 (initial trees = 10, learning rate/shrinkage = 0.005, tree complexity = 4, maximum trees = 10,000). We produced predictions across each of the 500 bootstrapped

A – Training region



B – Evaluation region

C – MESS



Fig 2. Data included for modelling across the training and evaluation regions and corresponding multivariate environmental similarity surface (MESS). A: The data-set of occurrence points and polygons used for fitting the boosted regression tree model across the model training region of Malaysia, Brunei, and Singapore. Presence polygons are displayed as the number of polygons covering each given pixel, with this density being proportional to the probability distribution of points sampled from the polygons for each bootstrap. B: The presence and absence records used in the model evaluation process, across the evaluation region of Southeast Asia excluding Malaysia, Brunei and Singapore. C: Multivariate environmental similarity surface (MESS) for the model, where areas shaded in light grey indicate that at least one covariate value at that point is outside the range of values within the training data (extrapolation). Administrative boundary base maps sourced from the Malaria Atlas Project (CC BY 3.0, [32]) and international boundaries from the US Department of State Large Scale International Boundaries dataset (public domain, [33]).

https://doi.org/10.1371/journal.pntd.0011570.g002

models, with summary statistics including mean, variance, and interquartile range calculated for each 5×5 km grid cell across Southeast Asia (Figure A in <u>S1 Text</u>). Predictions were made using covariate data corresponding to 2019, the most recent year available. As in the 2015 model, we restricted predictions to areas within the range of macaque and mosquito species known to be required for zoonotic transmission of *P. knowlesi* (i.e. the overlap in range maps of at least one reservoir and one vector species), using predicted species extent maps previously reported [37].

We produced a multivariate environment similarity surface (MESS) map (Fig 2C), indicating geographic areas where the value of at least one environmental covariate was outside the range of values present in the training data (i.e. the model is extrapolating) or vice-versa [52]. Prediction results for each bootstrapped model, rasters of summary statistics, the code used to produce results, and the updated occurrence database have been made available at osf.io/ k5bsa (DOI 10.17605/OSF.IO/K5BSA).

Model evaluation

We evaluated the model's predictive performance by calculating the area under the curve (AUC) metric across both the training and evaluation datasets. For the training dataset, we estimated a 10-fold cross-validated AUC throughout the tree count optimisation process, and reported the training AUC for each bootstrap as that of the optimal model selected. Across the evaluation dataset, we calculated AUC across each bootstrapped model, with pairwise distance selection of samples performed to avoid spatial sorting bias [53].

We calculated covariate relative influence scores for each bootstrapped model, representing the number of times a variable is selected for regression tree splitting, weighted by the squared improvement to the model as a result of each split and averaged over all trees [54]. We summarised these scores across the models as means and 95% confidence intervals, with mean values also being used to rank the relative covariate importance. We further calculated accumulated local effect (ALE) scores to describe the average effect of a covariate on the prediction value across the range of each covariate. The ALE score achieves this by identifying how the model prediction changes in response to small changes in the covariate of interest while all other covariates are kept constant, allowing for the effects of covariates to be identified even when the covariates may be highly correlated [55].

Results

Infection data

The literature review of articles including data on *P. knowlesi* infection occurrences published between October 2015 and March 2020 returned 511 candidate articles. Following a review of titles and abstracts, 159 articles were deemed likely to contain data for extraction, and 56 articles were identified as meeting the final criteria (Fig 1A). From these 56 articles, 264 occurrences of *P. knowlesi* were extracted, with 91 (34%) being assigned a point record type and 173 (66%) being assigned a polygon record type. Of the 264 extracted records, 241 (91%) were infections identified in humans, with only 14 in macaques and nine in mosquitoes (Table A in S1 Text). The number of records by year of sample collection was greatest in 2014 with 80 records across 14 publications (Fig 1C).

A majority of records added to the 2015 database were collected in Malaysia (n = 201, 76%). Within Malaysia, the spatial distribution of records was highly heterogeneous (Fig 1, Table A in <u>S1 Text</u>), with 127 polygon records assigned to the region of Sabah in contrast to the three records identified in the capital region of Kuala Lumpur. Malaysia was also the location of eight of the nine observed infected mosquitoes in the dataset, consistent with the greater sampling effort within the country [19].

Our literature search reveals that more infection occurrences from Indonesia have been reported since 2015, comprising 17% (n = 45) of the new presence records (where the prior 2015 literature search identified only five infections within the country). These records are the result of a small number of high-quality surveys and case reports from Aceh [56] and North Sumatra [57]. The literature review dataset contains three records from Laos, where the first confirmed human *P. knowlesi* infection was reported in 2016 [58].

In combination with the 260 occurrences used in the 2015 analysis, the total number of infection occurrences used in model fitting and evaluation was 524. Of these, 396 were within

the training region of Malaysia, Brunei and Singapore, with the remaining 128 located elsewhere in Southeast Asia (Table A in <u>S1 Text</u>).

Transmission suitability model output

The mean and standard deviation of predicted *P. knowlesi* transmission suitability (a relative measure of the potential risk of *P. knowlesi* transmission to humans) across at-risk areas of Southeast Asia is presented in Fig 3. Predictions were produced using covariate raster datasets as of 2019, representing our most up-to-date estimate of transmission suitability across the region. Further summary statistics of transmission suitability are presented in Figure A in <u>S1</u> Text.

The map of *P. knowlesi* transmission suitability (Fig 3A) shows highly heterogeneous levels of predicted risk across Southeast Asia. On the island of Borneo, all areas other than lowerlying coastal regions are expected to have a relatively high risk of *P. knowlesi* transmission. Other more sparsely distributed areas of relatively high risk are predicted in Indonesia within the provinces of Sumatra and West Nusa Tenggara. Peninsula Malaysia is predicted to have inland areas of high transmission risk. Thailand, Laos, Cambodia, Vietnam, Myanmar and the Philippine island of Luzon have smaller, localised areas of high predicted risk, with greater uncertainty in these predictions (Fig 3B) as a result of environmental differences to the model training region of Malaysia, Brunei and Singapore.

Within the training region, a mean area under the curve (AUC) of 0.81 was produced across the 500 bootstrapped models with a standard error of 0.001. For the evaluation region, the mean AUC was found to be 0.75 with a standard error of 0.003.



Fig 3. Predicted *P. knowlesi* **transmission suitability across Southeast Asia. A:** Modelled transmission suitability mean over Southeast Asia across the 500 bootstraps. Results are displayed only where an area is within the range of both a vector and reservoir species necessary for transmission (see <u>Methods</u>), regions outside of this range (displayed as grey) are considered to be very low risk for *P. knowlesi* transmission. Transmission suitability is a relative measure of the risk of *P. knowlesi* transmission from known reservoir species (via vector species) to humans. **B:** Standard deviation of the predicted transmission suitability across the 500 bootstraps. Administrative boundary base maps sourced from the Malaria Atlas Project (<u>CC BY 3.0, [32]</u>) and international boundaries from the US Department of State Large Scale International Boundaries dataset (<u>public domain, [33]</u>).

https://doi.org/10.1371/journal.pntd.0011570.g003

These values indicate a high degree of predictive performance.

Examining predictions within the evaluation region of the model (Southeast Asia excluding Malaysia, Brunei and Singapore), we may qualitatively assess the model's predictive performance. Regions with both a high modelled transmission suitability and previously identified occurrence samples of *P. knowlesi*—indicative of good model sensitivity—include the Aceh province of Sumatra island in Indonesia, the Koh Kong province in southern Cambodia and the Mimaropa region of the Philippines (Figure B in S1 Text, panel A). We also see that there are a substantial number of regions where the model predicts high transmission suitability where *P. knowlesi* occurrence has not previously been identified as of the 2020 literature review (i.e. omission errors [59]). This includes the province of West Nusa Tenggara in Indonesia (Figure B in S1 Text, panel A). Such predictions may be suggestive of a lack of surveillance in these regions, or that an environment is conducive to transmission but currently lacking wide-spread occurrence of a necessary vector or host species.

We additionally examine the performance of the predicted *P. knowlesi* transmission risk map presented by the 2015 analysis against the occurrence data collected in our literature review (Figure G in <u>S1 Text</u>). We find that, qualitatively speaking, the performance of the 2015 analysis in predicting the presence of infection occurrences published in the literature between October 2015 and March 2020 was good.

The covariate of human population density was found to have the highest ranked relative influence for the majority (496/500, 99.2%) of the bootstrapped models, closely followed by that of healthcare accessibility. Mean and 95% confidence intervals of relative influence scores across bootstraps are presented in Figure E in <u>S1 Text</u>. The new covariates of tree coverage and forest loss were found to be highly influential; out of 21 covariates (20 environmental covariates and the species covariate), the median rank for tree coverage was 5 (95% confidence interval: 2–11), and for forest loss was 10 (95% confidence interval: 6–15). Plots of the accumulated local effects (ALE) describing the influence of each continuous covariate across the covariate's range are presented in Figure F in <u>S1 Text</u>.

Discussion

In this study, we utilised an environmental niche modelling approach to predict the relative suitability for *P. knowlesi* transmission to humans across Southeast Asia. We extended a previous analysis that incorporated data up to 2015 [31] by adding infection and environmental data up to 2020, and improving the utilisation of data on land use patterns. Through a review of literature published between October 2015 and March 2020, we identified 264 published occurrences of *P. knowlesi*. This resulted in a total of 524 records being utilised in model fitting and evaluation for the current study. As changes in *P. knowlesi* transmission risk may be expected where substantial amounts of deforestation have occurred [24–27], we now capture this in the model by deriving annual forest loss and coverage datasets. We predict that the distribution of *P. knowlesi* risk is highly heterogeneous across Southeast Asia, with the largest areas of predicted risk in Malaysia and Indonesia, and smaller, localised regions of high risk predicted in the Greater Mekong Subregion, The Philippines and Northeast India.

Our analysis can help to guide the prioritisation of locations for future sampling and surveillance for P. knowlesi malaria by highlighting areas of high predicted risk that may have been under-sampled. Since the publication of the 2015 analysis, there has been no change to the World Health Organization's malaria elimination status of any country believed to be at risk for indigenous *P. knowlesi* transmission [60]. However, within the Greater Mekong Subregion of Cambodia, Myanmar, Thailand, Laos and Vietnam, substantial declines have been observed in the total number of reported malaria cases as of 2021 [61]. The 2015 analysis

noted that Laos, Myanmar, Thailand and Vietnam were likely high-value sites for future sampling efforts [31] and our literature search revealed only a small number of additional *P. knowlesi* occurrences in these countries as of 2020 (Fig 1, Table A in <u>S1 Text</u>). Our current analysis predicts localised areas of moderate-to-high relative transmission risk in this region (Fig 3), suggesting an ongoing need for surveillance of *P. knowlesi* malaria. In contrast, Palawan in The Philippines was also highlighted as a target for future sampling efforts in the 2015 analysis [31], with new occurrence records confirming the presence of *P. knowlesi* in this region [62].

Indonesia has a stated goal to eliminate malaria by 2030 [63,64], and may be on track given that a majority of administrative regions have declared elimination [65]. However, the presence of *P. knowlesi* across the country presents a serious challenge to these efforts. In March 2022, the WHO Malaria Policy Advisory Group (MPAG) concluded that certification of malaria elimination status should only occur where the risk of *P. knowlesi* was 'negligible', i.e. below some low threshold of annual incidence [60,66]–a requirement that has already prevented Malaysia from receiving elimination certification [67]. Given this requirement, continued surveillance and mitigation of *P. knowlesi* throughout at-risk regions of Indonesia will be important. Between the period of 2015 and 2020, a small number of studies have identified substantial numbers of *P. knowlesi* infections within Indonesia [68–70], particularly within northern Sumatra [56,71,72], a region identified as a valuable target for surveillance effort in the 2015 model [31]. Despite this, the Indonesian region of Kalimantan on the island of Borneo still has a relative scarcity of occurrence data given its high predicted transmission suitability and the number of P. knowlesi cases reported in adjacent areas of Malaysia.

Our map of *P. knowlesi* transmission risk may also help to quantitatively guide site selection for public health surveillance or intervention. For example, surveillance sampling could be concentrated in regions where the model predicts a high transmission suitability but with a high variance, such that the understanding of the geographical distribution of the disease is maximised for the least effort and that the uncertainty in these regions could be reduced in future risk mapping outputs. If value was instead placed on maximising the probability of identifying cases of *P. knowlesi*, sampling could be concentrated where a high transmission suitability is accompanied by lower variance. Efficient deployment of sampling resources could be achieved by combining the modelling outputs with constraints; for example, sites where access would require a prohibitive amount of travel time could be excluded [73].

Updating a model of risk—as we have performed here for *P. knowlesi* transmission suitability-raises key questions regarding when and why such an update should be performed. We produced the update in response to two primary factors: the accumulation of further P. know*lesi* infection occurrence data since the publication of the previous mapping study, which was expected to improve estimate precision when incorporated into the model; and changes in land cover across Southeast Asia, such as deforestation, which were suspected to have caused changes in the underlying distribution of transmission risk. For our predictions (with covariate data as of 2019 and occurrence data up to 2018) to be sufficient for future sampling efforts, the change in the underlying distribution of risk over time should be minimal. However, identifying if such changes have occurred is particularly difficult for *P. knowlesi* given the nature of data collection for the pathogen. Whereas studies of the human malaria species are able to isolate the effect of a changing risk distribution through the use of data which has been collected in a systematic manner [74], the data available for our study largely comprises infection occurrences identified by localised prospective sampling and passive surveillance. Despite these difficulties, we note that the good predictive performance of the 2015 analysis (Figure G in S1 Text) provides reason to believe that our static estimates are sufficient to inform future sampling efforts.

As niche modelling frameworks are correlative, the secondary results described in this work should be interpreted with care. The relative influence scores (Figure E in <u>S1 Text</u>) and the accumulated local effect plots (Figure F in <u>S1 Text</u>) may provide insight into risk factors for *P. knowlesi* transmission. However, these results do not provide evidence for causal relationships, which would instead be more appropriately identified through studies utilising a causal inference framework. For example, the covariate of healthcare accessibility, which ranks highly according to relative influence scores (Figure E in <u>S1 Text</u>), could capture a direct causal effect on the risk of being diagnosed with *P. knowlesi* (e.g. likelihood that someone is identified as having a *P. knowlesi* infection increasing with access to healthcare) or may simply be confounded by a common variable (e.g. likelihood of acquiring a *P. knowlesi* infection increasing for those who work at plantations, confounded by such plantations occurring in areas of lower healthcare accessibility).

Our model predicts the relative suitability for *P. knowlesi* transmission, not the prevalence of infection nor the incidence of cases (which would require different input data that are not widely available for *P. knowlesi* malaria). While transmission suitability is a useful metric for prioritising locations for future P. knowlesi surveys, the absolute values are specific to the input data and model parameterisation, and we therefore cannot directly compare absolute values produced by the model presented here and those from the model developed in 2015. Although we expect that the transmission suitability prediction produced by either of the models should be qualitatively related to the underlying 'true' risk of *P. knowlesi* infection, little can be said of this relationship other than that it is expected to be monotonic under the assumption that the background data points are biased in the same manner as the presence data [50]. This means, for instance, that any differences between the models that could arise as a result of dilation in this relationship (such as the upwards dilation observed in Figure D in S1 Text) cannot be taken alone as indicating a change in underlying transmission suitability.

Noting the limitations in these comparisons, we find that the predictions in our work and the 2015 model broadly align, though with clear differences in the local spatial variation of the prediction surface (Figure C in S1 Text, Figure D in S1 Text, panel A). As an example, on the island of Borneo our predictions form a smooth region of high predicted risk, whereas in the 2015 model predictions over the same area varied substantially at a small spatial scale; this pattern is repeated similarly elsewhere across Southeast Asia [31]. In countries such as Laos, Myanmar and Vietnam, we predict overall a lower transmission suitability than those presented in the 2015 model, though within these countries we continue to predict small areas of high transmission risk. Comparing the overall distributions of predicted transmission suitability between the 2015 and 2020 models shows that our new predictions produce a more highly contrasting bimodal distribution of risk compared to that produced by the 2015 model [31] (Figure D in <u>S1 Text</u>, panel A). Identifying potential reasons for such changes in model output is also difficult. Such an analysis would require a systematic examination of differences in model structure, covariate data and occurrence data, and, given the correlative nature of the model, would not necessarily be expected to provide insight into the mechanisms of P. knowlesi risk.

The temperature suitability index covariate used in the model attempts to describe the effect of temperature on the basic reproduction number for some combination of malaria parasites and mosquito vectors. As data on the incubation periods for *P. knowlesi* under differing temperatures and mosquito hosts is currently unavailable, no suitability index for the species can currently be produced. In this work, we instead utilise a proxy in the form of a suitability index for *P. falciparum* [42]. Even if this proxy does not itself accurately capture mechanistic limits on *P. knowlesi* reproduction, it is not immediately obvious what bias this would introduce into the results, if any, as the boosted regression tree model may still infer suitability under some

transformation of the index. There is clear value in further laboratory research on the reproduction of *P. knowlesi* under different temperatures that could inform a species-specific suitability index.

It is believed that workers involved in the development and cultivation of oil palm plantations are at greater risk for developing *P. knowlesi* infection given their proximity to *P. knowlesi* vector and reservoir species [75]. However, we were unable to include this as a covariate in our model as there is currently no published dataset of palm oil plantations with complete coverage across the Southeast Asia region.

In our approach, we implicitly assume that under-ascertainment of *P. knowlesi* infections in humans due to asymptomatic/submicroscopic or spontaneously resolving disease (see [9-11]) has a uniform effect geographically. If the data used for this study were biased by such under-ascertainment which was not uniform across space (e.g. due to differing levels of immunity between regions), this would be expected to in turn bias our predicted transmission suitability downwards in environments similar to those where under-ascertainment were occurring. Further research on any potential spatial association of asymptomatic or submicroscopic human *P. knowlesi* infection would be of high value in further refining estimates of the spatial distribution of transmission of the parasite.

Annual data is not available for some of the covariates used in the model where the underlying phenomena may be expected to change over time; the covariates of reservoir/vector species distribution and temperature suitability are dependent upon variables such as climate or land cover, and the covariate of healthcare accessibility is dependent upon changes in transportation infrastructure and locations of healthcare sites. In lieu of available data on change in these covariates over time they are instead assumed to be constant. In effect, this means that the modelled species distributions as of 2014, temperature suitability index for *P. falciparum* as of 2010 and healthcare accessibility as of 2019 are all assumed constant over the years 2001 to 2019. Future modelling efforts could be improved by considering the change in these covariates over time.

Our map of *P. knowlesi* transmission suitability predict high *P. knowlesi* disease risk across broad areas of Southeast Asia, with large regions of high predicted *P. knowlesi* risk that have not yet been sampled for the pathogen. Our work demonstrates the importance of continued surveillance and prospective sampling of the pathogen, especially in regions where malaria elimination is currently being pursued.

Supporting information

S1 Text. SupplementaryMaterials—Updating estimates of Plasmodium knowlesi malaria risk in response to changing land use patterns across Southeast Asia. Figure A in S1 Text. Summary statistics for modelled transmission suitability across Southeast Asia, calculated across the set of 500 bootstraps. Results are only displayed where an area is in the range of both a vector and reservoir species necessary for transmission (see Methods). Administrative boundary base maps sourced from the Malaria Atlas Project (CC BY 3.0, [32]) and international boundaries from the US Department of State Large Scale International Boundaries dataset (public domain, [33]). Figure B in S1 Text. Predicted mean transmission suitability with overlay of infection occurrence data across the evaluation region. Results are only displayed where an area is in the range of both a vector and reservoir species necessary for transmission (see Methods). A: Predicted transmission suitability with infection occurrence polygons and points in blue. B: Predicted transmission suitability with infection absence polygons in blue. Administrative boundary base maps sourced from the US Department of State Large Scale International Boundaries from the US Department of State Large Scale International Boundaries and points in blue. B: Predicted transmission suitability with infection occurrence polygons and points in blue. B: Predicted transmission suitability with infection absence polygons in blue. Administrative boundary base maps sourced from the Malaria Atlas Project (CC BY 3.0, [32]) and international boundaries from the US Department of State Large Scale International Boundaries dataset (public domain, [33]). Figure C in S1 Text. Comparison of the modelled

mean transmission suitability value between the current work and the 2015 model. Model predictions are presented from the current work with data as of 2020 (A), and the predictions presented in the 2015 model [31] (B). Note that the absolute value of predictions are not necessarily comparable given differences in model specification and training data. Results are only displayed where an area is in the range of both a vector and reservoir species necessary for transmission (see Methods). Administrative boundary base maps sourced from the Malaria Atlas Project (CC BY 3.0, [32]) and international boundaries from the US Department of State Large Scale International Boundaries dataset (public domain, [33]). Figure D in S1 Text. Changes in the distribution of predicted transmission suitability between the current study and the 2015 model. Histograms (A) and quantile-quantile plot (B) comparing the distributions of mean predicted transmission suitability for the 2015 and 2020 models of P. knowlesi transmission risk. Histograms are presented on a relative x-axis (ranging from minimal to maximal predicted mean risk), with quartiles of predicted risk displayed as dashed vertical lines. Figure E in S1 Text. Relative influence scores for each covariate. Scores are calculated for each bootstrap, with points and lines representing median values and 95% confidence intervals respectively. Figure F in S1 Text. Accumulated local effect (ALE) plots for each covariate. ALE indicates the mean effect of changing a covariate's value upon the prediction (on logistic scale) across the range of that covariate. The ALE values are calculated for each bootstrap, with the median value, 50% and 95% confidence intervals presented as lines, darker shaded regions and lighter shaded regions respectively. Figure G in S1 Text. Comparison of predicted transmission risk as presented in the 2016 work and the occurrence data collected in the 2020 literature review. Results are only displayed where an area is in the range of both a vector and reservoir species necessary for transmission (see Methods). Administrative boundary base maps sourced from the Malaria Atlas Project (CC BY 3.0, [32]) and international boundaries from the US Department of State Large Scale International Boundaries dataset (public domain, [33]). Table A in S1 Text. The number of human, macaque and mosquito samples in the occurrence database produced by the 2015 literature review. Samples in Indonesia and Malaysia are shown stratified by region (province, state or territory). Total counts are shown for records from both the 2020 literature review and 2015 literature review.

(PDF)

Acknowledgments

This research was supported by The University of Melbourne's Research Computing Services and the Petascale Campus Initiative (www.unimelb.edu.au). We thank Dr Timothy William for their support. We would like to also thank the Director General of Health Malaysia for the permission to publish this article.

Author Contributions

Conceptualization: Matthew J. Grigg, Jennifer A. Flegg, David J. Price, Freya M. Shearer.

Data curation: Ruarai J. Tobin, Meg K. Tully.

Formal analysis: Ruarai J. Tobin, Lucinda E. Harrison.

Funding acquisition: Inke N. D. Lubis, Rintis Noviyanti, Nicholas M. Anstey, Giri S. Rajahram, Matthew J. Grigg, Jennifer A. Flegg, David J. Price, Freya M. Shearer.

Investigation: Ruarai J. Tobin.

- Methodology: Ruarai J. Tobin, Lucinda E. Harrison, Jennifer A. Flegg, David J. Price, Freya M. Shearer.
- Project administration: Inke N. D. Lubis, Rintis Noviyanti, Nicholas M. Anstey, Giri S. Rajahram, Matthew J. Grigg, Jennifer A. Flegg, David J. Price, Freya M. Shearer.
- **Resources:** Inke N. D. Lubis, Rintis Noviyanti, Nicholas M. Anstey, Giri S. Rajahram, Matthew J. Grigg, Jennifer A. Flegg, Freya M. Shearer.

Software: Ruarai J. Tobin, Lucinda E. Harrison.

Supervision: Jennifer A. Flegg, David J. Price, Freya M. Shearer.

Validation: Ruarai J. Tobin.

Visualization: Ruarai J. Tobin.

- Writing original draft: Ruarai J. Tobin, Freya M. Shearer.
- Writing review & editing: Ruarai J. Tobin, Lucinda E. Harrison, Meg K. Tully, Inke N. D. Lubis, Rintis Noviyanti, Nicholas M. Anstey, Giri S. Rajahram, Matthew J. Grigg, Jennifer A. Flegg, David J. Price.

References

- Vythilingam I, Wong ML, Wan-Yussof WS. Current status of Plasmodium knowlesi vectors: a public health concern? Parasitology. 2016; 145(1):32–40. https://doi.org/10.1017/S0031182016000901 PMID: 27222102
- Collins WE, Barnwell JW. Plasmodium knowlesi: Finally Being Recognized. The Journal of Infectious Diseases. 2009; 199(8):1107–1108. https://doi.org/10.1086/597415 PMID: 19284287
- Chin W, Alpert E, Collins WE, Jeter MH, Contacos PG. Experimental Mosquito-Transmission of Plasmodium Knowlesi to Man and Monkey. The American Journal of Tropical Medicine and Hygiene. 1968; 17(3):355–358. https://doi.org/10.4269/ajtmh.1968.17.355 PMID: 4385130
- Cuenca PR, Key S, Lindblade KA, Vythilingam I, Drakeley C, Fornace K. Is there evidence of sustained human-mosquito-human transmission of the zoonotic malaria Plasmodium knowlesi? A systematic literature review. Malaria Journal. 2022; 21(1). https://doi.org/10.1186/s12936-022-04110-z PMID: 35300703
- Singh B, Daneshvar C. Human Infections and Detection of Plasmodium knowlesi. Clinical Microbiology Reviews. 2013; 26(2):165–184. https://doi.org/10.1128/CMR.00079-12 PMID: 23554413
- 6. Coatney GR. The primate malarias. US National Institute of Allergy and Infectious Diseases; 1971.
- Brock PM, Fornace KM, Parmiter M, Cox J, Drakeley CJ, Ferguson HM, et al. Plasmodium knowlesi transmission: integrating quantitative approaches from epidemiology and ecology to understand malaria as a zoonosis. Parasitology. 2016; 143(4):389–400. https://doi.org/10.1017/ S0031182015001821 PMID: 26817785
- Barber BE, Grigg MJ, Cooper DJ, van Schalkwyk DA, William T, Rajahram GS, et al. Clinical management of Plasmodium knowlesi malaria. In: Current research on naturally transmitted Plasmodium knowlesi. Elsevier; 2021. p. 45–76. Available from: https://doi.org/10.1016/bs.apar.2021.08.004 PMID: 34620385
- Imwong M, Madmanee W, Suwannasin K, Kunasol C, Peto TJ, Tripura R, et al. Asymptomatic Natural Human Infections With the Simian Malaria Parasites Plasmodium cynomolgi and Plasmodium knowlesi. The Journal of Infectious Diseases. 2018; 219(5):695–702. <u>https://doi.org/10.1093/infdis/jiy519</u> PMID: 30295822
- Fornace KM, Nuin NA, Betson M, Grigg MJ, William T, Anstey NM, et al. Asymptomatic and Submicroscopic Carriage of Plasmodium knowlesi Malaria in Household and Community Members of Clinical Cases in Sabah, Malaysia. Journal of Infectious Diseases. 2015; 213(5):784–787. <u>https://doi.org/10.1093/infdis/jiv475</u> PMID: 26433222
- 11. Grigg MJ, Cox J, William T, Jelip J, Fornace KM, Brock PM, et al. Individual-level factors associated with the risk of acquiring human Plasmodium knowlesi malaria in Malaysia: a case-control study. The Lancet Planetary Health. 2017; 1(3):e97–e104. https://doi.org/10.1016/S2542-5196(17)30031-1 PMID: 28758162
- Daneshvar C, Davis TME, Cox-Singh J, Rafa'ee MZ, Zakaria SK, Divis PCS, et al. Clinical and Laboratory Features of human Plasmodium knowlesi infection. Clinical Infectious Diseases. 2009; 49(6):852–860. https://doi.org/10.1086/605439 PMID: 19635025

- Grigg MJ, William T, Barber BE, Rajahram GS, Menon J, Schimann E, et al. Age-Related Clinical Spectrum of Plasmodium knowlesi Malaria and Predictors of Severity. Clinical Infectious Diseases. 2018; 67 (3):350–359. https://doi.org/10.1093/cid/ciy065 PMID: 29873683
- Fornace KM, Herman LS, Abidin TR, Chua TH, Daim S, Lorenzo PJ, et al. Exposure and infection to Plasmodium knowlesi in case study communities in Northern Sabah, Malaysia and Palawan, The Philippines. PLOS Neglected Tropical Diseases. 2018; 12(6):e0006432. <u>https://doi.org/10.1371/journal.pntd.</u> 0006432 PMID: 29902171
- Fornace KM, Brock PM, Abidin TR, Grignard L, Herman LS, Chua TH, et al. Environmental risk factors and exposure to the zoonotic malaria parasite Plasmodium knowlesi across northern Sabah, Malaysia: a population-based cross-sectional survey. The Lancet Planetary Health. 2019; 3(4):e179–e186. https://doi.org/10.1016/S2542-5196(19)30045-2 PMID: 31029229
- Vadivelan M, Dutta T, et al. Recent advances in the management of Plasmodium knowlesi infection. Trop Parasitol. 2014; 4(1):31–4. https://doi.org/10.4103/2229-5070.129158 PMID: 24754024
- Mahittikorn A, Masangkay FR, Kotepui KU, Milanez GDJ, Kotepui M. Quantification of the misidentification of Plasmodium knowlesi as Plasmodium malariae by microscopy: an analysis of 1569 P. knowlesi cases. Malaria Journal. 2021; 20(1). https://doi.org/10.1186/s12936-021-03714-1 PMID: 33836773
- Lee KS, Cox-Singh J, Singh B. Morphological features and differential counts of Plasmodium knowlesi parasites in naturally acquired human infections. Malaria Journal. 2009; 8(1). https://doi.org/10.1186/ 1475-2875-8-73 PMID: 19383118
- Cooper DJ, Rajahram GS, William T, Jelip J, Mohammad R, Benedict J, et al. Plasmodium knowlesi Malaria in Sabah, Malaysia, 2015–2017: Ongoing Increase in Incidence Despite Near-elimination of the Human-only Plasmodium Species. Clinical Infectious Diseases. 2019; 70(3):361–367. https://doi.org/ 10.1093/cid/ciz237 PMID: 30889244
- Tyagi RK, Das MK, Singh SS, Sharma YD. Discordance in drug resistance-associated mutation patterns in marker genes of Plasmodium falciparum and Plasmodium knowlesi during coinfections. J Antimicrob Chemother. 2013; 68(5):1081–1088. https://doi.org/10.1093/jac/dks508 PMID: 23292346
- 21. Organization WH. Global technical strategy for malaria 2016–2030. World Health Organization; 2015.
- Muh F, Kim N, Nyunt MH, Firdaus ER, Han JH, Hoque MR, et al. Cross-species reactivity of antibodies against Plasmodium vivax blood-stage antigens to Plasmodium knowlesi. PLOS Neglected Tropical Diseases. 2020; 14(6):e0008323. https://doi.org/10.1371/journal.pntd.0008323 PMID: 32559186
- Chin AZ, Maluda MCM, Jelip J, Jeffree MSB, Culleton R, Ahmed K. Malaria elimination in Malaysia and the rising threat of Plasmodium knowlesi. Journal of Physiological Anthropology. 2020; 39(1). https:// doi.org/10.1186/s40101-020-00247-5 PMID: 33228775
- Cuenca PR, Key S, Jumail A, Surendra H, Ferguson HM, Drakeley CJ, et al. Epidemiology of the zoonotic malaria Plasmodium knowlesi in changing landscapes. In: Current research on naturally transmitted Plasmodium knowlesi. Elsevier; 2021. p. 225–286. Available from: https://doi.org/10.1016/bs.apar. 2021.08.006 PMID: 34620384
- Brock PM, Fornace KM, Grigg MJ, Anstey NM, William T, Cox J, et al. Predictive analysis across spatial scales links zoonotic malaria to deforestation. Proceedings of the Royal Society B: Biological Sciences. 2019; 286(1894):20182351. https://doi.org/10.1098/rspb.2018.2351 PMID: 30963872
- Davidson G, Chua TH, Cook A, Speldewinde P, Weinstein P. The Role of Ecological Linkage Mechanisms in Plasmodium knowlesi Transmission and Spread. EcoHealth. 2019; 16(4):594–610. <u>https://doi.org/10.1007/s10393-019-01395-6 PMID</u>: 30675676
- Stark DJ, Fornace KM, Brock PM, Abidin TR, Gilhooly L, Jalius C, et al. Long-Tailed Macaque Response to Deforestation in a Plasmodium knowlesi-Endemic Area. EcoHealth. 2019; 16(4):638–646. https://doi.org/10.1007/s10393-019-01403-9 PMID: 30927165
- Reisen WK. Landscape Epidemiology of Vector-Borne Diseases. Annual Review of Entomology. 2010; 55(1):461–483. https://doi.org/10.1146/annurev-ento-112408-085419 PMID: 19737082
- Yu H, Liu X, Kong B, Li R, Wang G. Landscape ecology development supported by geospatial technologies: A review. Ecological Informatics. 2019; 51:185–192. https://doi.org/10.1016/j.ecoinf.2019.03.006
- Anderson RP, Mart´ınez-Meyer E, Nakamura M, Arau´jo MB, Peterson AT, Sober´on J, et al. Ecological Niches and Geographic Distributions (MPB-49). Princeton University Press; 2011. https://doi.org/10. 1515/9781400840670
- Shearer FM, Huang Z, Weiss DJ, Wiebe A, Gibson HS, Battle KE, et al. Estimating geographical variation in the risk of zoonotic Plasmodium knowlesi infection in countries eliminating malaria. PLOS Neglected Tropical Diseases. 2016; 10(8):e0004915. https://doi.org/10.1371/journal.pntd.0004915 PMID: 27494405
- Pfeffer DA, Lucas TCD, May D, Harris J, Rozier J, Twohig KA, et al. malariaAtlas: an R interface to global malariometric data hosted by the Malaria Atlas Project. Malar J. 2018; 17(1):352. <u>https://doi.org/ 10.1186/s12936-018-2500-5 PMID: 30290815</u>

- The Office of the Geographer and Global Issues at the U S Department of State. Large Scale International Boundaries (LSIB);. <u>https://data.humdata.org/</u> dataset/large-scale-international-boundarieslsib.
- Weiss DJ, Nelson A, Gibson HS, Temperley W, Peedell S, Lieber A, et al. A global map of travel time to cities to assess inequalities in accessibility in 2015. Nature. 2018; 553(7688):333–336. <u>https://doi.org/ 10.1038/nature25181 PMID: 29320477</u>
- Weiss DJ, Nelson A, Vargas-Ruiz CA, Gligori c K, Bavadekar S, Gabrilovich E, et al. Global maps of travel time to healthcare facilities. Nature Medicine. 2020; 26(12):1835–1838. https://doi.org/10.1038/ s41591-020-1059-1 PMID: 32989313
- Potapov P, Yaroshenko A, Turubanova S, Dubinin M, Laestadius L, Thies C, et al. Mapping the World's intact forest landscapes by remote sensing. Ecology and Society. 2008; 13(2).
- Moyes CL, Shearer FM, Huang Z, Wiebe A, Gibson HS, Nijman V, et al. Predicting the geographical distributions of the macaque hosts and mosquito vectors of Plasmodium knowlesi malaria in forested and non-forested areas. Parasites & vectors. 2016; 9(1):1–12. <u>https://doi.org/10.1186/s13071-016-1527-0</u> PMID: 27125995
- Hansen MC, Potapov PV, Moore R, Hancher M, Turubanova SA, Tyukavina A, et al. High-resolution global maps of 21st-century forest cover change. science. 2013; 342(6160):850–853. <u>https://doi.org/ 10.1126/science.1244693</u> PMID: 24233722
- Farr TG, Rosen PA, Caro E, Crippen R, Duren R, Hensley S, et al. The shuttle radar topography mission. Reviews of geophysics. 2007; 45(2).
- Lobser SE, Cohen WB. MODIS tasselled cap: land cover characteristics expressed through transformed MODIS data. International Journal of Remote Sensing. 2007; 28:5079–5101.
- Weiss DJ, Atkinson PM, Bhatt S, Mappin B, Hay SI, Gething PW. An effective approach for gap-filling continental scale remotely sensed time-series. ISPRS Journal of Photogrammetry and Remote Sensing. 2014; 98:106–118. https://doi.org/10.1016/j.isprsjprs.2014.10.001 PMID: 25642100
- **42.** Gething PW, Van Boeckel TP, Smith DL, Guerra CA, Patil AP, Snow RW, et al. Modelling the global constraints of temperature on transmission of Plasmodium falciparum and P. vivax. Parasites & vectors. 2011; 4(1):1–11. https://doi.org/10.1186/1756-3305-4-92 PMID: 21615906
- Gaughan AE, Stevens FR, Linard C, Jia P, Tatem AJ. High Resolution Population Distribution Maps for Southeast Asia in 2010 and 2015. PLOS ONE. 2013; 8(2):e55882. <u>https://doi.org/10.1371/journal.pone.0055882</u> PMID: 23418469
- Gaughan AE, Stevens FR, Linard C, Jia P, Tatem AJ. High resolution population distribution maps for Southeast Asia in 2010 and 2015. PLOS one. 2013; 8(2):e55882. <u>https://doi.org/10.1371/journal.pone.</u> 0055882 PMID: 23418469
- 45. Friedl MA, Sulla-Menashe D, Tan B, Schneider A, Ramankutty N, Sibley A, et al. MODIS Collection 5 global land cover: Algorithm refinements and characterization of new datasets. Remote sensing of Environment. 2010; 114(1):168–182.
- De'ath G, Fabricius KE. Classification and regression trees: A powerful yet simple technique for ecological data analysis. Ecology. 2000; 81(11):3178–3192. https://doi.org/10.1890/0012-9658(2000)081
- 47. Elith J, Leathwick JR, Hastie T. A working guide to boosted regression trees. Journal of Animal Ecology. 2008; 77(4):802–813. https://doi.org/10.1111/j.1365-2656.2008.01390.x PMID: 18397250
- Elith J, Graham CH, Anderson RP, Dud´ık M, Ferrier S, Guisan A, et al. Novel methods improve prediction of species' distributions from occurrence data. Ecography. 2006; 29(2):129–151. https://doi.org/10.1111/j.2006.0906-7590.04596.x
- Leathwick J, Elith J, Francis M, Hastie T, Taylor P. Variation in demersal fish species richness in the oceans surrounding New Zealand: an analysis using boosted regression trees. Marine Ecology Progress Series. 2006; 321:267–281. https://doi.org/10.3354/meps321267
- Phillips SJ, Dud´ık M, Elith J, Graham CH, Lehmann A, Leathwick J, et al. Sample selection bias and presence-only distribution models: implications for background and pseudo-absence data. Ecological Applications. 2009; 19(1):181–197. https://doi.org/10.1890/07-2153.1 PMID: 19323182
- Groves C. Primate Taxonomy. Smithsonian Series in Comparative Evolutionary Biology. Washington, D.C., DC: Smithsonian Books; 2001.
- Elith J, Kearney M, Phillips S. The art of modelling range-shifting species. Methods in Ecology and Evolution. 2010; 1(4):330–342. https://doi.org/10.1111/j.2041-210x.2010.00036.x
- Hijmans RJ. Cross-validation of species distribution models: removing spatial sorting bias and calibration with a null model. Ecology. 2012; 93(3):679–688. https://doi.org/10.1890/11-0826.1 PMID: 22624221
- 54. Friedman JH. Greedy function approximation: a gradient boosting machine. Annals of statistics. 2001; p. 1189–1232.

- Molnar C. Interpretable Machine Learning: A Guide For Making Black Box Models Explainable. Independently published; 2020.
- 56. Herdiana H, Cotter C, Coutrier FN, Zarlinda I, Zelman BW, Tirta YK, et al. Malaria risk factor assessment using active and passive surveillance data from Aceh Besar, Indonesia, a low endemic, malaria elimination setting with Plasmodium knowlesi, Plasmodium vivax, and Plasmodium falciparum. Malaria journal. 2016; 15(1):1–15. https://doi.org/10.1186/s12936-016-1523-z PMID: 27619000
- Lubis IND, Wijaya H, Lubis M, Lubis CP, Divis PCS, Beshir KB, et al. Contribution of Plasmodium knowlesi to Multispecies Human Malaria Infections in North Sumatera, Indonesia. The Journal of Infectious Diseases. 2017; 215(7):1148–1155. https://doi.org/10.1093/infdis/jix091 PMID: 28201638
- Iwagami M, Nakatsu M, Khattignavong P, Soundala P, Lorphachan L, Keomalaphet S, et al. First case of human infection with Plasmodium knowlesi in Laos; 2018.
- 59. Peterson AT. Mapping disease transmission risk: enriching models using biogeography and ecology. JHU Press; 2014.
- 60. World Health Organization. World malaria report 2022. 2022;.
- 61. World Health Organization. Accelerating malaria elimination in the Greater Mekong. 2022;.
- Gamalo LE, Dimalibot J, Kadir KA, Singh B, Paller VG. Plasmodium knowlesi and other malaria parasites in long-tailed macaques from the Philippines. Malar J. 2019; 18(1):147. <u>https://doi.org/10.1186/</u> s12936-019-2780-4 PMID: 31014342
- Departemen Kesehatan. Keputusan Menteri Kesehatan Republik Indonesia Nomor 293/MENKES/SK/ IV/2009 28 April 2009 tentang Eliminasi Malaria di Indonesia; 2009.
- 64. Ministry of Health Indonesia–Kementerian Kesehatan Republik Indonesia. Challenges Toward Malaria Elimination 2030. 2021;.
- Sitohang V, Sariwati E, Fajariyani SB, Hwang D, Kurnia B, Hapsari RK, et al. Malaria elimination in Indonesia: halfway there. The Lancet Global Health. 2018; 6(6):e604–e606. https://doi.org/10.1016/S2214-109X(18)30198-0 PMID: 29703663
- Organisation WH, et al. WHO Malaria Policy Advisory Group (MPAG) meeting: meeting report, March. 2022;.
- Fornace KM, Zorello Laporta G, Vythilingham I, Chua TH, Ahmed K, Jeyaprakasam NK, et al. Simian malaria: a narrative review on emergence, epidemiology and threat to global malaria elimination. Lancet Infect Dis. 2023; https://doi.org/10.1016/S1473-3099(23)00298-0 PMID: 37454671
- Lubis IND, Wijaya H, Lubis M, Lubis CP, Sutherland CJ. Molecular identification of human Plasmodium knowlesi infections in North Sumatera, Indonesia. International Journal of Infectious Diseases. 2016; 45:182. https://doi.org/10.1016/j.ijid.2016.02.425
- 69. Wibowo AA, Umniyati SR, Hutagalung J, Rahayu T. Confirmation of Anopheles balabacensis as natural vector of malaria caused by Plasmodium knowlesi inhabits forested areas in Kecamatan Balik Bukit, Western Lampung Regency. E3S Web of Conferences. 2020; 151:01028. <u>https://doi.org/10.1051/e3sconf/202015101028</u>
- Setiadi W, Sudoyo H, Trimarsanto H, Sihite BA, Saragih RJ, Juliawaty R, et al. A zoonotic human infection with simian malaria, Plasmodium knowlesi, in Central Kalimantan, Indonesia. Malaria Journal. 2016; 15(1). https://doi.org/10.1186/s12936-016-1272-z PMID: 27083152
- Herdiana H, Irnawati I, Coutrier FN, Munthe A, Mardiati M, Yuniarti T, et al. Two clusters of Plasmodium knowlesi cases in a malaria elimination area, Sabang Municipality, Aceh, Indonesia. Malaria journal. 2018; 17(1):1–10.
- Coutrier F, Cotter C, Tirta YK, Schwartz A, Zarlinda I, Basri HH, et al. Serial molecular identification to confirm the presence of Plasmodium knowlesi in Indonesia. In: American Journal of Tropical Medicine and Hygiene. vol. 93; 2015. p. 263–263.
- Longbottom J, Krause A, Torr SJ, Stanton MC. Quantifying geographic accessibility to improve efficiency of entomological monitoring. PLOS Neglected Tropical Diseases. 2020; 14(3):e0008096. https://doi.org/10.1371/journal.pntd.0008096 PMID: 32203517
- 74. Hay SI, Snow RW. The Malaria Atlas Project: developing global maps of malaria risk. PLoS Med. 2006; 3(12):e473. https://doi.org/10.1371/journal.pmed.0030473 PMID: 17147467
- 75. Naserrudin NA, Hod R, Jeffree MS, Ahmed K, Culleton R, Hassan MR. The Role of Human Behavior in Plasmodium knowlesi Malaria Infection: A Systematic Review. International Journal of Environmental Research and Public Health. 2022; 19(6):3675. <u>https://doi.org/10.3390/ijerph19063675</u> PMID: 35329359